Author's response to reviews

Title: Pylorus resection or pylorus preservation in partial pancreatico-duodenectomy (PROPP study): study protocol for a randomized controlled trial.

Authors:

Thilo Hackert (thilo.hackert@med.uni-heidelberg.de)
Thomas Bruckner (bruckner@imbi.uni-heidelberg.de)
Colette Doerr-Harim (colette.doerr-harim@med.uni-heidelberg.de)
Phillip Knebel (phillip.knebel@med.uni-heidelberg.de)
Markus K. Diener (markus.diener@med.uni-heidelberg.de)
Werner Hartwig (werner.hartwig@med.uni-heidelberg.de)
Oliver Strobel (oliver.strobel@med.uni-heidelberg.de)
Stefan Fritz (stefan.fritz@med.uni-heidelberg.de)
Lutz Schneider (lutz.schneider@med.uni-heidelberg.de)
Jens Werner (jens.werner@med.uni-heidelberg.de)
Markus W. Buechler (markus.buechler@med.uni-heidelberg.de)

Version: 4 Date: 21 January 2013

Author's response to reviews: see over
Dear editors,

Thank you very much for the re-review of our manuscript “Pylorus resection or pylorus preservation in partial pancreato-duodenectomy (PROPP study): study protocol of a randomized controlled trial.”

We have inserted one sentence in the “trial status” with the date of enrolment of the first patient which we have just performed. The change has been highlighted in the text.

We would be happy if the revised manuscript could be considered for publication in TRIALS.

On behalf of all authors,

yours sincerely.

Thilo Hackert